Author’s response to reviews

Title: How organisations promoting vaccination respond to misinformation on social media: a qualitative investigation

Authors:

Maryke Steffens (maryke.steffens@hdr.mq.edu.au)
Adam Dunn (adam.dunn@mq.edu.au)
Kerrie Wiley (kerrie.wiley@sydney.edu.au)
Julie Leask (julie.leask@sydney.edu.au)

Version: 4 Date: 14 Sep 2019

Author’s response to reviews:

15 September 2019

Authors’ replies to Editor's comments for: How organisations promoting vaccination respond to misinformation on social media: a qualitative investigation

Authors: Maryke S. Steffens, Adam G. Dunn, Kerrie E. Wiley, Julie Leask

Thank you for your correspondence requesting revisions for our paper. We have addressed each of requested revisions below.

1. Keywords

This section has been moved to the end of the Abstract.

2. Declarations

This section has been moved to after the Conclusions section.
Warm regards,

Maryke Steffens